Author's Response To Reviewer Comments

Dr Hans Zauner Assistant Editor GigaScience

We double checked the manuscript and perform minor corrections.

We also moved Figure S1 (ipê photo) to the main-text (it is now Figure 1) and change figure and supplemental material numbers.

Hope our changes are satisfactory.

All the best Rosane Collevatti